Author's response to reviews

Title: Variations in rates of nosocomial infection among Canadian neonatal intensive care units may be practice-related

Authors:

Khalid Aziz (kaziz@mun.ca)
Douglas D McMillan (Douglas.McMillan@iwk.nshealth.ca)
Wayne Andrews (wandrews@morgan.ucsmun.ca)
Margaret Pendray (mpendray@cw.bc.ca)
Zhenguo Qiu (winston_qiu@yahoo.ca)
Shoo K Lee (shool@interchange.ubc.ca)
Canadian NEONATAL Network (hbavinton@cw.bc.ca)

Version: 2 Date: 18 January 2005

Author's response to reviews:

I have re-uploaded the manuscript, tables and figures as some of the tables and figures were missing after the first submission.